Author's response to reviews

Title: Normative data and discriminative properties of short form 36 (SF-36) in Turkish urban population

Authors:

Yucel Demiral (yucel.demiral@deu.edu.tr)
Gul Ergor (gul.ergor@deu.edu.tr)
Belgin Unal (belgin.unal@deu.edu.tr)
Semih Semin (semih.semin@deu.edu.tr)
Yildiz Akvardar (yildiz.akvardar@deu.edu.tr)
Berna Kivircik (berna.kivircik@deu.edu.tr)
Koksal Alptekin (koksal.alptekin@deu.edu.tr)

Version: 7 Date: 7 October 2006

Author's response to reviews:

Dear Editor,
Thank you for the revision notes for the MS (# 1005918185992970). We have made the following changes you have kindly requested.
Regards
Belgin Unal

1. The following statement has been added to the end of author's contributions section. (page 13, lines 8-9)

"All authors read and approved the final manuscript."

2. Asterisks have been removed from top of the authors' names in the title page since there is no "equal" contribution among the authors. Contributions of the authors have been indicated clearly in the "author's contributions section". This section is designed according to Journal style. Page 13.

"YD designed the study, organized the data collection and performed the data analysis; he drafted and revised the manuscript. GE participated in the initial study design and revision of the article. BU participated in data analysis and revision of the manuscript. SS participated in the initial study design, collection of data and revision of the manuscript. YA, BK and KA were participated in the initial study design, and revising the manuscript. All authors read and approved the final version of this manuscript."

3. The keyword section have removed from the manuscript.